Author’s response to reviews

Title: Significant proportion of acute hepatitis B in Poland in 2010-2014 attributed to hospital transmission: combining surveillance and public registries data.

Authors:
Małgorzata Stępień (mstepien@pzh.gov.pl)
Karolina Zakrzewska (kzakrzewska@pzh.gov.pl)
Magdalena Rosinska (mrosinska@pzh.gov.pl)

Version: 3 Date: 20 Mar 2018

Author’s response to reviews:
Dear Editor,

Thank you for your additional comments, the answers are given below:

1) This study was supported by the National Institute of Public Health – National Institute of Hygiene under the statutory task, no external funds was obtained. In the section ‘Declarations, Funding’ (Line #334, page 15) the following sentence have been added:

"This study was supported by the National Institute of Public Health – National Institute of Hygiene under the task no. 6/EM.1/2017".

2) In the previous version of the manuscript, tables 2 and 3 have been cited, but unsuitable shortcut 'Tab.' was used instead of ‘Table’, I apologize for this error.

"Table 2" was cited in lines 172-173, page 8 and "Table 3" in lines 180 and 181, page 9.

Corrections have been made in the manuscript and clean version of the manuscript was uploaded.
Regards,

Małgorzata Stępień